Author's response to reviews

Title: Mediating Effects of Job Satisfaction and Moderating Effects of Role Stress on Nursing Personnel's Organizational Commitment in Taiwan's Medical Centers

Authors:

Hsin Hsin Chang (easyhhc@mail.ncku.edu.tw)
Ching Sheng Chang (arthur0660@yahoo.com.tw)

Version: 5 Date: 11 August 2007

Author's response to reviews: see over
1. **Title of article**: “Mediating Effects of Job Satisfaction and Moderating Effects of Role Stress on Nursing Personnel’s Organizational Commitment in Taiwan’s Medical Centers”

2. **Authors**: Hsin Hsin Chang and Ching Sheng Chang

3. **Contact**: Ching Sheng Chang

4. **Affiliation**: Dept. of Business Administration, National Cheng Kung University

5. **E-mail**: arthur0660@yahoo.com.tw

6. **Postal Address**: Dept. of Business Administration, National Cheng Kung University, 1 University Rd., Tainan City 70101, Taiwan

7. **Tel. & Fax**: +886-7-5584262